The Effects of Macroprudential Policies on Research and Development Expenditure

A System GMM Approach



Nina Maria Lancelot & Charlie Nilsson

Supervisor: Marta Giagheddu

Department of Economics

Master Thesis I, May 2022

Abstract

Preliminary firm level evidence suggests that macroprudential policies reduce firm credit growth and MSME investment, and could therefore have a negative impact on economic growth. However, the question remains if macroprudential policies affect R&D investments as well. Although R&D is primarily financed internally, the literature also suggests that debt financing may be used to some degree depending on firm and country characteristics. Furthermore, implementation of macroprudential policies may also affect R&D indirectly by reducing access to external finance for other investments, compelling firms to deprioritize R&D in their use of internal funds. This study is a first attempt to investigate the effects of borrower–targeted macroprudential policies on research and development expenditures. We estimate a dynamic panel model of R&D financing using a system GMM approach on a heterogeneous sample of European firm level data. Our results indicate that implementation and tightening of borrower–targeted macroprudential policies is associated with lower R&D expenditures for MSMEs and unlisted firms. No evidence is found that these policies affect large and listed firms.

Keywords: macroprudential policy, research and development, economic growth, financial constraints

Contents

1	Intr	oductio	on		1				
2 Previous Literature									
	2.1	Macrop	prudential Policies		3				
	2.2	R&D F	Financing		5				
	2.3	Theore	etical Framework		8				
3	Dat	a			10				
4	Met	hod			12				
	4.1	The G	MM System Estimator		12				
	4.2	Specify	ving the Model		14				
5	Res	d Analysis		17					
	5.1	Results	s for All Firms		17				
5.2 Results for Size Classifications									
	5.3	Results	s for Listing Status		20				
	5.4	Analys	is of the Results		22				
		5.4.1	Implications of the Results		23				
	5.5	Robust	tness Checks		24				
		5.5.1	Comparison to Brown et al. (2012)		24				
		5.5.2	Weighting the Sample		25				
6	Con	clusion	1		28				
7	App	oendix			33				

List of Tables

3.1	Sample Summary Statistics	11
5.1	Results for All Firms	17
5.2	Results for SMEs and Large Firms	19
5.3	Results for Unlisted and Listed Firms	20
5.4	Comparison of Weighted and Unweighted Results, BOR1	26
5.5	Comparison of Weighted and Unweighted Results, BOR2	27
7.1	Specification of Brown et al.	33
7.2	Countries Included in the Full Sample	34

1 Introduction

Following the Global Financial Crisis of 2008, policies that seek to enhance system-wide financial stability rapidly became a topic of interest for both policymakers and academics. These macroprudential policies have in effect existed since the 1930's, but had up until 2008 mostly been used in developing economies. Since then, more advanced economies have implemented a broad variety of instruments that the literature has not yet extensively analyzed (Galati and Moessner, 2013). Although efforts have been made to reduce this knowledge gap, the literature is primarily focused on whether the various instruments are effective in achieving their primary goal to reduce procyclicality and structural risks. Relatively little focus has been devoted to investigating their potential externalities.

A welcome exception is the small but growing body of literature on macroprudential policies and growth. However, there is still no consensus among authors on whether there is a positive or negative relationship between the two (Ma, 2020). While some evidence suggests that active use of macroprudential policies enhance growth by reducing economic volatility (Boar et al., 2017), other evidence suggests that they have a negative effect on average growth by restricting economic activity (Sánchez and Röhn, 2016). A study which provides insight of through which channels macroprudential policies can affect growth is a recent paper by Ayyagari et al. (2018). The paper demonstrates that especially borrower-targeted instruments reduce firm credit growth, resulting in a reduction in sales growth and investments of micro, small and medium-sized enterprises (MSMEs).

Since it has been documented that macroprudential policies could negatively affect firm investments, a natural addition to the literature is to investigate which *types* of investments are most affected. Of particular relevance to the relationship between macroprudential policies and growth is whether investment in research and development (R&D) is affected, considering that such activities are an integral part of the growth process. Although R&D is often primarily funded internally (e.g., Hall and Lerner, 2010), evidence suggests that added availability of external finance increases R&D expenditures for smaller firms (Czarnitzki and Hottenrott, 2009) and that bank funding of R&D is especially common in some countries (e.g., Bhagat and Welch, 1995). Moreover, it is possible that macroprudential policies affect R&D indirectly by reducing access to external finance for other investments, compelling firms to deprioritize R&D in their use of internal funds.

This paper contributes to the existing literature on both macroprudential policies and R&D financing by presenting preliminary evidence of the relationship between the two. We empirically analyze data of 2,078 European firms, which differ in both size and listing status, using a system GMM approach. We build upon the dynamic panel model developed by Brown et al. (2012) by including two measures of macroprudential policy implementation and estimating their effects on R&D expenditures. The model is applied to different subsets of our sample, investigating the effects of macroprudential policies on R&D expenditures depending on firm size and listing status. Our results indicate that implementation and tightening of borrower-targeted macroprudential policies are associated with lower R&D expenditures for small and unlisted firms. We find no evidence of these policies affecting the expenditures of large or listed firms.

The structure of the paper proceeds as follows: Section 2 reviews previous research on the effects of macroprudential policies and on R&D financing sources, as well as presents the theoretical framework of this paper. Section 3 presents an overview of the data used in the study. Section 4 explains the methodology of system GMM estimation and presents the specification of our model. Section 5 presents the results, our analysis and two robustness checks. Section 6 concludes the paper and offers suggestions for further research.

2 Previous Literature

2.1 Macroprudential Policies

The purpose of macroprudential policies is to reduce system-wide financial risk, such as interbank exposures and cyclical volatility in credit and asset prices. Since the Global Financial Crisis of 2008, several such policy instruments have been implemented in advanced economies. These can be categorized as either lender– or borrower–targeted, depending on whether they regulate the supply– or demand–side of credit. Common examples of borrower-targeted instruments are loan–to–value (LTV) ratios and debt–to– income (DTI) ratios, which apply to the demand side of credit. Respectively, examples of lender–targeted instruments are caps on foreign currency lending, reserve requirements and ceilings on credit or credit growth, all of which apply to the supply–side of credit (Galati and Moessner, 2013).

As previously mentioned, the body of literature on macroprudential policy instruments is still quite small, although efforts have been made to reduce the knowledge gap. Among the first empirical approaches is Lim et al. (2011), in which the authors perform a crosscountry regression analysis of data from 49 countries during 2000-2010 to study the effects of ten macroprudential policy instruments on financial risk. Lim et al. include both borrower and lender targeted instruments in their model, and find that almost all policies (including DTI and LTV ratios) were effective in dampening procyclicality. In another influential study, Cerutti et al. (2017) reach a similar conclusion using an Arellano and Bond (1991) GMM approach on data from 119 countries during 2000-2013. Their results suggest that macroprudential policies are generally associated with a lower credit growth rate for both households and the corporate sector, although most notably for households and in developing and closed economies.

The empirical literature thus suggests that macroprudential policies have been effective in their primary objective to increase financial stability, which in turn has more recently raised questions about their potential impact on economic growth. The reason for this is that studies have shown that other stability-enhancing policies increase growth (e.g., Aghion et al., 2012), indicating that macroprudential policies may have the same effect. One of the first accounts to empirically investigate any potential link between macroprudential policies and growth was Sánchez and Röhn (2016). The authors find that limits on DTI ratios and taxes on financial institutions are negatively correlated with extreme positive and negative growth shocks, while limits on LTV ratios, limits on interbank exposures and taxes on financial institutions are only negatively correlated with extreme positive growth shocks. The authors also find that limits on LTV ratios, limits on interbank exposures and taxes on financial institutions are correlated with lower growth on average (Sánchez and Röhn, 2016). The evidence thus implies a trade-off between stability and growth in the use of such instruments. Conversely, Boar et al. (2017) find no evidence of a trade-off. Using a dynamic GMM approach on panel data for 64 countries over a five-year period, the authors present evidence that intensive use of macroeconomic policies are associated with both higher and less volatile GDP per capita growth rates.

One of the channels through which macroprudential policies can negatively affect growth is through the limitation of financial resources (Boar et al., 2017). This aspect is captured by Ayyagari et al. (2018), who use firm level data to capture the effects of macroprudential policies on firm credit growth and investment. Using data on 900 000 firms in 48 countries during 2003-2011, Ayyagari et al. show that macroeconomic policies reduce firm credit growth, as a consequence of decreased access to bank loans. The magnitude of the effect varied depending on firm characteristics, with smaller and younger firms being relatively more affected. Furthermore, the results were found to be more pronounced when considering borrower-targeted instruments than lender-targeted. Additionally, Ayyagari et al. show that MSMEs experience relatively lower investment and sales growth after implementation of borrower-targeted instruments. The authors contend that these results imply that MSMEs use bank loans as a source of finance for their investments, and that implementation of macroprudential policies reduces access to this debt. This effect is only prevalent for MSMEs because of these firms being more bank dependent and financially constrained, meaning that it is costly and difficult for these firms to fund their activities. Larger firms typically have greater access to both external equity and internal financing, making them less sensitive to such policy changes (Ayyagari et al., 2018).

2.2 R&D Financing

As for all types of investments, firms can finance their R&D activities using internal funds (such as revenue), external funds (such as bank loans) or a combination of the two. Although external financing is an alternative, it has long been recognized that R&D activities have several characteristics that make them more challenging to finance using external sources compared to other investments (Czarnitzki and Hottenrott, 2009). One of these characteristics is that both the main input and output of R&D activities is knowledge, which is an intangible asset. A large part of all R&D expenditures are wages and salaries paid to highly skilled employees, and the main output of their efforts is firm-specific knowledge on how to produce new goods and services (Hall and Lerner, 2010). Therefore, R&D activities rarely involve assets that can be used as collateral for loans, which increases the risk of lending for potential creditors (Müller and Zimmermann, 2006). Moreover, since this intangible asset is partly embedded in the employees, it is lost if they leave the firm. Therefore, firms tend to smooth their R&D expenditures over time in order to avoid lay-offs. R&D spending thus exhibits high adjustment costs, and the required rate of return to such an investment can be high in order to cover them (Hall and Lerner, 2010). Another distinguishing characteristic is the high degree of uncertainty regarding the output value of R&D activities. R&D is often a high risk – high reward type of investment, and the quality of a proposed project can be difficult to evaluate for lenders due to the often highly technical nature of such projects (Ughetto, 2008). This property generates a problem of asymmetric information, which is exacerbated by the fact that firms often wish to keep their research projects as secret as possible due to the risk of imitating competitors. Furthermore, even if a R&D project is successful, lenders are unable to participate in the potentially large rewards since external credit typically comes at a fixed, predetermined interest rate.

Because of the above-listed reasons, access to external financing is typically believed to be more restrictive for R&D than for other types of investments. Consequentially, internal financing is expected to be the preferred option (Howe and McFetridge, 1976; Wang et al., 2016). With the background of internal funds being the primary source of R&D financing, the literature on R&D financing has devoted much attention to detecting constraints due to lacking internal funds (Czarnitzki and Hottenrott, 2009). A common approach to empirically test for financial constraints is to estimate the sensitivity of R&D to cash flow, a method originally developed by Fazzari et al. (1988). The rationale behind this approach is that if a firm responds to an increase in liquidity by increasing investments, there must have been investment opportunities available to the firm before the increase which it chose not to act on. In other words, the firm is assumed not to have been liquid enough to undertake the investment before the increase, and either have chosen to refrain from using external capital or been unable to do so. In these situations, the investment decision is made subject to financial constraints (Hall and Lerner, 2010).

However, as pointed out in Brown et al. (2012), the empirical literature on R&D cash flow sensitivities contains highly mixed results. For example, it has often been noted that there are differences in cash flow sensitivities across firms of different sizes and maturity. Small and young firms have been shown to typically exhibit greater cash flow sensitivities than large and mature firms. Harhoff (1998) presents evidence of such a difference for a diverse sample of German firms during 1990–1994. Using an accelerator model, an errorcorrection model and a dynamic panel model derived from a Euler equation for investment, Harhoff finds that small firms exhibit cash flow sensitivities, but that such an effect is weak or non-existent for large firms. Similarly, Brown et al. (2012) estimate these effects using a dynamic panel model on a large sample of European firms during 1995–2007. Brown et al. find that even when controlling for R&D smoothing and revenues from issuing external equity, cash flow sensitivities were only detected for relatively small firms in the sample. A viable explanation for differences in cash flow sensitivities across firm size and maturity is that the above-mentioned challenges of external finance for R&D investment are exacerbated for small and young firms. These groups typically have less assets to offer as collateral, information asymmetries are generally larger, and there is a heightened risk of default for young firms. An implication of the aforementioned circumstances is that these firms have limited access to bank loans (Czarnitzki and Hottenrott, 2009). The results are also consistent with the more general literature on investment theory, which, as mentioned in the previous section, suggests that small and young firms are generally more financially constrained (Ayyagari et al., 2018).

Furthermore, differences in cash flow sensitivities have also been noted across countries, or more specifically, across financial market regimes. The literature often focuses on differences between market– and bank–based economies. Countries including the United States, the United Kingdom and Sweden are characterized as being market–based, implying highly developed stock markets and advanced venture capital industries. Conversely, bank-based countries such as Italy, Germany and France have a larger focus on debt financing and bank intermediation (Hall and Lerner, 2010). Because of the use of more external financing in market-based economies, firms in these countries have been shown to be more sensitive to fluctuations in cash flow. For example, Bond et al. (2005) use a dynamic panel model on data from large manufacturing firms in the U.K. and Germany and find that while U.K. firms exhibit cash flow sensitivities, German firms did not. Similarly, Mulkay et al. (2001) use an error–correction model on data from large manufacturing firms in France and the U.S. and find that cash flow sensitivities were much larger in France. Across the dimension of financial market regimes, differences of firm size and age persist. Ughetto (2008) uses a difference–GMM approach on a diverse sample of Italian firms and finds that the cash flow–investment elasticity is significantly higher for small firms than medium-sized and large. One interpretation of these results is that firms in bank–based economies are less financially constrained because they have access to bank loans, while firms in market-based economies do not (Czarnitzki and Hottenrott, 2009).

An alternative strategy used for studying R&D financing is to investigate the relationship between firms' financial structure and R&D performance. Although differing in methodology, the results of such studies are highly similar to those that estimate cash flow sensitivities. For example, Bhagat and Welch (1995) provide evidence of this using a cross-sectional VAR method on a sample of American, Canadian, European and Japanese firms during 1985–1990. Bhagat and Welch find that leverage has a positive impact on R&D expenditures in Japan, which is another bank-based economy, but a negative impact in the market–based United States. Similarly, Lucey and Bhaird (2006) find no relationship between leverage and R&D intensity for MSMEs in Ireland. Much like in the literature on cash flow sensitivities, these results imply that firms in bank-based economies have larger access to bank loans for their R&D activities. Moreover, the literature suggests that the more R&D intensive a firm is, the less it is leveraged. For example, Aghion et al. (2004) find that leverage decreases as R&D intensity increases for a sample of listed U.K. firms. Evidence of the relationship between financial structures and R&D shows that all though leverage decreases with R&D intensity, R&D intensive firms are, in fact, leveraged. The results thus indicate that firms may use debt as a source of finance for their R&D, and at the very least show that R&D performing firms use debt for other

activities. Furthermore, the literature suggests that this is not only true for large firms or firms in bank-based countries, but for MSMEs and firms in market-based economies as well. Further evidence in support of this can be found in Czarnitzki and Hottenrott (2009), in which the authors use a measure of credit rating to show that access to external funds has a positive impact on R&D intensity for German firms, especially smaller and younger ones.

Moreover, capital structure is related to another important part of R&D financing – the role of external equity. As is the case with debt, issuing equity is costly and tends to be associated with problems of asymmetric information when considering R&D activities (Hall and Lerner, 2010). However, raising funds through issuance of equity can have several advantages over debt; firms do not have to use assets as collateral and investors can share the potentially large profits from a successful project (Brown et al., 2012). Consequently, recent accounts have provided evidence that issuing equity can be an important part of R&D financing, especially for small and young firms. Brown et al. (2012) show that net proceeds from stock issuance have a large and significant impact on R&D expenditures in his sample of European firms, especially so for young and relatively small firms. Similarly, Magri (2014) provides evidence that issuing equity significantly increases the probability that firms invest in R&D, most notably for small, young, and heavily leveraged firms.

2.3 Theoretical Framework

The purpose of this paper is to investigate the effects of macroprudential policies on R&D expenditures. To the best of our knowledge, no similar study has yet been produced on this topic, and so there is little theoretical precedent for a potential relationship between the two. However, as can be understood from the separate bodies of literature on macroprudential policies and R&D financing, it is evident that: (i) borrower-targeted macroprudential policies reduce both firm credit growth and investment, (ii) although firms primarily rely on internal finance for their R&D investments, firms also make use of debt finance either specifically for their R&D investments or for other activities, and (iii), the use of external finance appears to vary across firm size, maturity and country. The combination of these results suggests that there could be some degree to which macroprudential policies affect R&D investment. However, this effect may be smaller for R&D expenditure than other types of investments due to debt being a disfavored source of financing for R&D activities. We propose that the impact can be either direct or indirect, depending on whether debt finance is used specifically for R&D activities or for other investments. If indirect, use of macroprudential policies imposes or intensifies financial constraints on the investment that debt is used for, thereby compelling firms to substitute use of debt with internal funds. Since internal funds are limited, firms will need to prioritize their use of them and possibly choose to reduce R&D spending.

Building upon the results in Ayyagari et al. (2018) and Czarnitzki and Hottenrott (2009), we expect that this relationship primarily holds for younger and smaller firms, and only for borrower-targeted macroprudential policies. As argued by Ayyagari et al. (2018), these firms are already financially constrained, and restricting their access to one source of finance would therefore exacerbate this circumstance. Since Czarnitzki and Hottenrott (2009) show that improving the access to debt has the largest effect on R&D expenditures for small and young firms, we suspect the opposite will hold in the case of restrictions. However, due to limited data availability, the scope of this study is restricted to differences in size and not maturity. Furthermore, the results of Ayyagari et al. were most conclusive for the borrower-targeted instruments LTV and DSTI ratios, which is expected since these instruments directly correspond to the firms' supply of assets and stream of cash flow. Consequentially, this paper will be restricted to the study of these specific instruments.

Additionally, we suspect that the effects of macroprudential policy implementations on R&D expenditures will be larger for firms that are not listed on a stock market than those that are. As discussed in the previous section, evidence suggests that issuing equity can be an important part of R&D financing, especially for smaller and younger firms. Therefore, firms that raise funds through equity are expected to be less dependent on bank loans than those that are not.

3 Data

Our sample consists of data of 2,078 firms from 14 countries from the European Union and the United Kingdom between 2013-2020.¹ The EU and the UK have been chosen for two reasons: (i) to estimate the effect of macroprudential policies on R&D expenditures, this study will largely build upon Brown et al. (2012) who made the same selection (ii) the countries in the sample have been relatively active in macroprudential policy implementation during this time period. The added limitation to the specific 14 countries has been made due to data availability, with countries who contributed less than 3 firms in the baseline regression having been removed. In accordance with previous literature on R&D financing, only manufacturing firms (with SIC codes between 2999-4000) have been included.² Furthermore, only firms which reported positive spending on research and development for 2016-2018 have been included. This restriction is partly due to Brown et al. (2012) only including firms with three consecutive years of data, but also to minimize gaps in the data.

Following Brown et al. (2012), several financial variables that determine R&D expenditures have been controlled for in the regression. Data on financial variables have been collected from Orbis (2022), a worldwide private company database. Besides R&D expenditures, these are: cash flow, cash holdings, operational revenue³ and stock issues. Cash flow is calculated gross of R&D expenditures and total depreciation, as specified in Brown et al.⁴ Cash holdings are measured as cash and equivalents at end of year. Stock issues denote the net proceeds from stock issuance each year. Variables have been scaled by total assets and winsorized at the 1% level, which is comparable to the 1% trimming of tails that was used in Brown et al. (2012). Summary statistics of the data are presented in Table 3.1.

¹A list of included countries and their number of firms can be found in the Appendix.

²Manufacturing firms have been shown to be the most research intense, and is the focus of the most literature on determinants of R&D expenditure (Brown et al., 2012; Ughetto, 2008; Hall, 1992; Hall and Lerner, 2001)

³Brown et al. controlled for sales and not operational revenue, but since many firms lack reporting of sales in our Orbis sample, operational revenue has been used as a proxy.

⁴Extraordinary items are not included in the cash flow calculation because of low data availability for the majority of firms.

Variable	Median	Mean	Observations					
Employees	233.00	5,018.19	27,704					
R&D Expenditure	0.02	0.05	23,630					
Operational Revenue	1.06	1.20	$27,\!367$					
Cash Holdings	0.08	0.14	$26,\!109$					
Stock Issues	0.00	0.05	4,779					
Cash Flow	0.13	0.14	$27,\!255$					
Index for Borrowing Targeted Policies	1.00	0.78	28,760					
Changes in Borrowing Targeted Policies	1.00	1.01	28,760					
All financial variables are scaled by total assets								

 Table 3.1: Sample Summary Statistics

Data on macroprudential policies have been collected from the IMF's integrated Macroprudential Policy (iMaPP) database, originally constructed by Alam et al. (2019). As previously described, only the borrower-targeted policies LTV and DSTI will be analyzed. LTV describes limits on loan-to-value ratios applied to residential and commercial mortgages. DSTI captures policies limiting the debt-service-to-income and loan-to-income ratio which restrict the size of debt service payments or a loan relative to income. Instead of estimating the individual impacts of these instruments, we follow Ayyagari et al. (2018) by constructing a borrower-targeted index containing information on both of them. This index will in turn be specified in two ways. The first is coded as 0, 1 or 2, indicating if none, either or both of the borrower-targeted instruments were in place in a specific country at a specific time. The second index is a measure of changes in borrower-targeted policies, and contains information on whether or not an instrument has become tightened (stricter) or loosened (less strict) in a specific year.

4 Method

4.1 The GMM System Estimator

To investigate the relationship between borrower-targeted policies and R&D expenditure, we build upon the dynamic panel model used in Brown et al. (2012). The general idea behind dynamic panel models is that there is serial correlation in the dependent variable, such that past values can be used as predictors for current values (Roodman, 2009). This assumption is reasonable when modeling R&D expenditures since they are believed to be highly persistent over time due to expenditure smoothing (Hall and Lerner, 2010). However, using lagged values of the dependent variable as predictors also implies some problems that should be addressed. The main concern is dynamic panel bias, emanating from the additional assumption that there are fixed individual effects in the model that are correlated with the dependent variable. Under this assumption, using lagged values of the dependent variable as regressors implies that there will be correlation between these regressors and the firm-specific effects. This violates the exogeneity condition, meaning that OLS estimators will be inconsistent for dynamic panel models. The problem of endogeneity still holds even when controlling for firm-specific effects using the Least Square Dummy Variable (LSDV) estimator (Roodman, 2009).

An approach which resolves these issues is system GMM estimaton, developed in Arellano and Bover (1995) and Blundell and Bond (1998). It builds on the approach by Arellano and Bond (1991), which is to transform the data to remove the individual fixed effects. The standard transformation is the first-difference one, in which the data matrices are left-multiplied with a matrix $I_N M_D$, where I_N is the identity matrix of order N (the number of individuals in the sample) and M_D is a diagonal matrix of -1s with a subdiagonal of 1s to the right. Performing this transformation removes the time invariant fixed effects in the error term since they are the same in each period (Roodman, 2009). This method can be illustrated with a simple model in which the lagged dependent variable is the only regressor and the error term consists of individual fixed effects and an idiosyncratic term:

$$y_{i,t} = \beta y_{i,t-1} + \varepsilon_{i,t}$$
$$\varepsilon_{i,t} = \alpha_i + v_{i,t}$$

After left-multiplying in matrix form, we obtain:

$$y_{i,t} - y_{i,t-1} = y_{i,t-1} - y_{i,t-2} + \alpha_i - \alpha_i + v_{i,t} - v_{i,t-1}$$
$$\Delta y_{i,t} = \beta \Delta y_{i,t-1} + \Delta v_{i,t}$$

While this transformation removes the individual fixed effects, the idiosyncratic error term and the lagged dependent variable are correlated through the t-1 terms. The solution to this problem is to use instruments for the lagged dependent variable. Conveniently, because of the assumption that there is serial correlation in the dependent variable, lagged values of the regressor constitute strong instruments while also eliminating the endogeneity (Roodman, 2009). Estimation is then performed using simple GMM. For the sake of simplicity, the above explanation considers an equation in which the lagged dependent variable is the only regressor in the model. The same principles still apply when adding more endogenous regressors.

Although the first-difference approach eliminates the individual fixed effects, it comes at the cost of magnifying gaps in the data. For example, if there is a missing value in an observation at t, using differences means that the observation will be dropped at both t and t - 1. Therefore, the differences approach exacerbates problems associated with an unbalanced panel (Roodman, 2009). Since the data used in this study produces an unbalanced panel, the alternative of orthogonal transformation will be used instead. This is another common approach which involves subtracting the average of all future available values instead of taking the first difference, such that:

$$\sqrt{T_{it}/(T_{it}+1)}(y_{i,t}-1/T_{it}\sum_{s < t} y_{i,s})$$

Where T_{it} is the number of available future observations and $\sqrt{T_{it}/(T_{it}+1)}$ is a scale factor. This strategy also removes the fixed effects since they are constant, meaning that average of all future fixed effects is equal to the contemporaneous fixed effect. However, as in the case with the first-difference transformation, there might still be endogeneity in the data due to inclusion of same-period values of the regressor and the idiosyncratic error term in the equation. Therefore, lagged values must still be used as instruments and estimation is done using GMM. Arellano and Bover (1995) and Blundell and Bond (1998) discovered efficiency can be increased further by combining the strategy of data-transformation with another one. Their idea was to purge fixed effects by first-differencing the *instruments* to make them exogenous, rather than first-differencing (or in another way transforming) the data. Instead of instrumenting the transformed variables with untransformed variables, Blundell and Bond instrument untransformed variables with transformed variables. This approach is valid under the additional assumption that first-differenced instruments are exogenous.

The two approaches are combined by adding the untransformed observations to transformed observations in each individual's data. This is done by left-multiplying the dataset by an augmented transformation matrix $M^+ = (M, I)'$ where M is the diagonal matrix used to transform the data. As for the instrument matrix, both transformed and untransformed instruments will now be included, and transformed instruments are used to estimate the untransformed observations and vice versa. Therefore, untransformed instruments are set to zero for untransformed observations and transformed instruments are set to zero for transformed observations (Roodman, 2009).

4.2 Specifying the Model

To explore the effects of macroprudential policies on R&D expenditure, we build upon the dynamic investment model developed by Brown et al. (2012). The model is based on the Euler equation for optimal capital accumulation under imperfect competition, which was originally presented by Bond and Meghir (1994). Their model is specified as:

$$Investment_{i,t} = \beta_1 Investment_{i,t-1} + \beta_2 Investment_{i,t-1}^2 + \beta_3 CashFlow_{i,t-1} + \beta_4 Output_{i,t-1} + \beta_5 Debt_{j,t-1}^2 + \beta_9 \Delta CashHoldings_{j,t} + d_t + \alpha_j + v_{j,t}$$
(1)

Where d_t is a time-specific effect, α_j is a firm-specific effect and $v_{j,t}$ is the idiosyncratic error term. All variables are scaled by capital stock. The main contribution of Brown et al. (2012) was to add contemporaneous and lagged cash holdings in order to control for the use of cash to smooth R&D expenditures over time, as discussed in the previous section. Furthermore, Brown et al. found no evidence of a significance of debt issuance in their sample, so the debt term was deleted from the model. Brown et al. also added Tobin's Q as an additional control for investment demand, proxied output with total sales and included contemporaneous versions of stock issuance and cash flow, leading to their final model:

$$RD_{j,t} = \beta_1 RD_{j,t-1} + \beta_2 RD_{j,t-1}^2 + \beta_3 Q_{j,t} + \beta_4 Sales_{j,t-1} + \beta_5 CashFlow_{j,t} + \beta_6 CashFlow_{j,t-1} + \beta_7 StkIssues_{j,t} + \beta_8 StkIssues_{j,t-1} + \beta_9 \Delta CashHoldings_{j,t} + \beta_{10} \Delta CashHoldings_{j,t-1} + d_t + \alpha_j + v_{j,t}$$

$$(2)$$

where all variables are scaled by beginning of term total assets. Equation (2) is the starting point for the main specification of this paper, and will be used as a baseline for R&D determinants.⁵ However, due to data availability, Tobin's Q is omitted from the final specification.⁶ Orbis data on this variable is mainly available for very large and listed companies, and this paper will consider slightly smaller firms and include both listed and unlisted firms. Similarly, when performing estimations for the entire sample, we will also omit stock issuance since only listed firms would be included in the regression otherwise. Stock issues are included again when we estimate the model for listed firms only.

Our extension of Equation (2) is to add borrower-targeted macroprudential policies. As described in Section 3, this variable will be specified in two different ways. In Equation (3), the variable is denoted by BOR and is specified as an index describing the number of borrower-targeted policies that were in place in a specific country in a given year. In equation (4), the variable denoted by Δ BOR2 indicates if a policy tightening or loosening has taken place in a given year. Following Ayyagari et al. (2018) the variable is lagged one period in both specifications, since the effect of the policies is expected to be delayed.

 $^{^{5}}$ As a robustness check, results will be compared to Brown et al. (2012) in Section 5.5 using more similar sample specification.

⁶Omitting Tobin's Q could have biased our results. To investigate this, we used sales growth as a proxy for Tobin's Q (which was also used as a robustness check in Brown et al., 2012) which did not have a significant impact on our results.

$$RD_{j,t} = \beta_1 RD_{j,t-1} + \beta_2 RD_{j,t-1}^2 + \beta_3 OpRevenue_{j,t-1} + \beta_4 CashFlow_{j,t} + \beta_5 CashFlow_{j,t-1} + \beta_6 StkIssues_{j,t} + \beta_7 StkIssues_{j,t-1} + \beta_8 \Delta CashHoldings_{j,t} + \beta_9 \Delta CashHoldings_{j,t-1} + \beta_{10} BOR_{j,t-1} + d_t + \alpha_j + v_{j,t}$$

$$(3)$$

$$\begin{split} RD_{j,t} = & \beta_1 RD_{j,t-1} + \beta_2 RD_{j,t-1}^2 + \beta_3 OpRevenue_{j,t-1} + \beta_4 CashFlow_{j,t} + \\ & \beta_5 CashFlow_{j,t-1} + \beta_6 StkIssues_{j,t} + \beta_7 StkIssues_{j,t-1} + \\ & \beta_8 \Delta CashHoldings_{j,t} + \beta_9 \Delta CashHoldings_{j,t-1} + \\ & \beta_{10} \Delta BOR2_{j,t-1} + d_t + \alpha_j + v_{j,t} \end{split}$$

$$\end{split}$$

$$\end{split}$$

$$\end{split}$$

$$\end{split}$$

Following Brown et al. (2012), all financial variables are assumed to be endogenous and are instrumented using their lagged values. This is a reasonable assumption given that operating revenue, cash flow, stock issuance and cash holdings should all be correlated with firm-specific effects. Unlike in Brown et al., 1–2 lags are used as instruments for the transformed equation and one period lags are used as instruments for the untransformed equation.⁷

Unlike the financial variables, the borrower-targeted macroprudential policy variables should not be correlated with firm-specific effects and are thus assumed to be exogenous. Therefore, these variables will instrument themselves and their coefficients are estimated using the IV-estimator.⁸ Robust standard errors are used to account for heteroscedasticity. Furthermore, year variables are included as dummies since the robustness and autocorrelation tests assume no correlation across individuals in the idiosyncratic disturbances, and the dummies control for the correlation that is expected to prevail (Roodman, 2009).

⁷Brown et al. (2012) uses lagged levels of t-3 and t-4 as instruments for the transformed regression, and lags dated t-4 for the untransformed regression.

⁸Instruments have been chosen according to results from the AR1, AR2 and Hansen–J tests. The instruments used are assumed to be valid when the p–value for the AR1 test is rejected and the AR2 and Hansen–J tests are not rejected.

5 Results and Analysis

5.1 Results for All Firms

The estimation results for all firms in our sample are reported in Table 5.1. Following Brown et al. (2012), variables have been sequentially added to the specification. Column 1 reports the estimation results without controlling for Δ CashHoldnings, and Column 2 reports the results when doing so. Our contribution – the results of adding macroprudential policy variables – are reported in columns 3 and 4 when controlling for all financial variables. This step-wise inclusion will allow us to compare the results of different specifications.

	(1)	(2)	(3)	(4)
RD_{t-1}	$\begin{array}{c} 0.635^{***} \\ (0.111) \end{array}$	$\begin{array}{c} 0.657^{***} \\ (0.131) \end{array}$	$\begin{array}{c} 0.659^{***} \\ (0.131) \end{array}$	$\begin{array}{c} 0.654^{***} \\ (0.132) \end{array}$
RD_{t-1}^2	-0.0525 (0.280)	0.0276 (0.284)	0.0219 (0.283)	0.0298 (0.285)
$OpRevenue_{t-1}$	-0.00423^{*} (0.00255)	-0.00492 (0.00305)	-0.00500 (0.00307)	-0.00497 (0.00305)
CashFlow	$\begin{array}{c} 0.0533^{***} \\ (0.00854) \end{array}$	$\begin{array}{c} 0.0595^{***} \\ (0.0105) \end{array}$	$\begin{array}{c} 0.0594^{***} \\ (0.0105) \end{array}$	$\begin{array}{c} 0.0597^{***} \\ (0.0105) \end{array}$
$\operatorname{CashFlow}_{t-1}$	-0.0170^{***} (0.00658)	-0.0158^{**} (0.00776)	-0.0159^{**} (0.00776)	-0.0156^{**} (0.00777)
$\triangle CashHoldings$		-0.0301^{***} (0.00595)	-0.0301^{***} (0.00594)	-0.0300^{***} (0.00596)
$\triangle \mathbf{CashHoldings}_{t-1}$		$\begin{array}{c} 0.00324 \\ (0.00637) \end{array}$	0.00329 (0.00637)	0.00327 (0.00637)
BOR_{t-1}			$\begin{array}{c} 0.000402 \\ (0.000993) \end{array}$	
$\triangle BOR2_{t-1}$				-0.00115 (0.00115)
Observations	11130	8973	8973	8973
Firms	2078	1982	1982	1982
No. of instruments	21	25	26	26
AR1 (p-value)	0.000	0.000	0.000	0.000
AR2 (p-value)	0.202	0.335	0.335	0.335
Hansen-J (p-value)	0.0623	0.0918	0.0926	0.0954

Table 5.1: Results for All Firms

Standard errors in parentheses

* p<0.10, * * $p{<}0.05, * * * p{<}0.010$

The coefficient for the lag of RD is relatively large and significant at the 1 percent level for every specification, and becomes slightly larger when including lagged and contemporaneous Δ CashHoldnings and BOR. The lag of RD² is insignificant in all specifications. The effect of lagged OpRevenue is significant at the 10 percent level in Column 1 and is insignificant in the other specifications. CashFlow is positive and significant at the 1 percent level for all specifications, with a slightly larger coefficient when including lagged and contemporaneous Δ CashHoldnings and larger so when including Δ BOR2. Lagged Cash-Flow is negative and small in absolute value with a 1 percent significance level in Column 1. In the specifications that include lagged and contemporaneous Δ CashHoldnings, it is significant at the 5 percent level with a slightly less negative coefficient. Δ CashHoldnings has a negative coefficient which is significant at the 1 percent level in all specifications where it is included, and remains largely unchanged by including BOR or Δ BOR2. The lag of Δ CashHoldnings is insignificant in all specifications. Neither BOR nor Δ BOR2 are significant at any conventional level, and coefficients of both variables are negligible.

5.2 Results for Size Classifications

Table 5.2 presents the estimation results when dividing the sample into categories according to size classification.⁹ As in 5.1, variables are added sequentially. The results for MSMEs are reported in columns 1–4 and follow a similar pattern to the results for all firms. The lag of RD continues to be significant at the 1 percent level, although becomes smaller when including lagged and contemporaneous $\Delta CashHoldnings$. The coefficients for lagged RD are smaller for MSMEs than for all firms (0.561 in Column 2 compared to 0.657). The lag of RD^2 continues to be insignificant. OpRevenue is significant at the 10 percent level for all MSME specifications, with negative coefficients which are small in absolute values. CashFlow is positive and significant at the 1 percent level in all specifications, and becomes slightly larger when controlling for lagged and contemporaneous Δ CashHoldnings. Lagged CashFlow continues to be negative, and becomes less significant (from the 5 to 10 percent level) when including cash holdings in the specification. Δ CashHoldnings is negative and significant at the 1 percent level, and has similar values in regression with and without controlling for BOR and Δ BOR2. The most important distinction from when estimating for all firms is that the coefficient for BOR is negative and significant at the 10 percent level. $\Delta BOR2$ remains insignificant in all specifications.

⁹MSMEs are defined as firms with 249 or fewer employees and large firms are defined as those with 250 or more employees. Firms which have not reported the number of employees are classified according to turnover based on the convention of the European Commission Communication.

	MSMEs				Large	Firms		
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)
RD_{t-1}	$\begin{array}{c} 0.646^{***} \\ (0.134) \end{array}$	$\begin{array}{c} 0.561^{***} \\ (0.153) \end{array}$	0.560^{***} (0.153)	0.560^{***} (0.152)	$\begin{array}{c} 0.770^{***} \\ (0.155) \end{array}$	0.896^{***} (0.169)	$\begin{array}{c} 0.898^{***} \\ (0.170) \end{array}$	0.896^{***} (0.168)
RD_{t-1}^2	-0.0670 (0.286)	0.243 (0.310)	0.243 (0.308)	$0.246 \\ (0.310)$	-0.380 (0.490)	-0.573 (0.431)	-0.577 (0.431)	-0.573 (0.430)
$OpRevenue_{t-1}$	-0.00546^{*} (0.00322)	-0.00596^{*} (0.00355)	-0.00593^{*} (0.00353)	-0.00597^{*} (0.00355)	-0.00395 (0.00374)	-0.00288 (0.00498)	-0.00301 (0.00503)	-0.00293 (0.00502)
CashFlow	$\begin{array}{c} 0.0523^{***} \\ (0.00996) \end{array}$	$\begin{array}{c} 0.0612^{***} \\ (0.0124) \end{array}$	$\begin{array}{c} 0.0612^{***} \\ (0.0124) \end{array}$	$\begin{array}{c} 0.0611^{***} \\ (0.0124) \end{array}$	$\begin{array}{c} 0.0323^{***} \\ (0.0113) \end{array}$	0.0330^{**} (0.0129)	0.0330^{**} (0.0129)	0.0331^{**} (0.0129)
$CashFlow_{t-1}$	-0.0170^{**} (0.00821)	-0.0177^{*} (0.0101)	-0.0173^{*} (0.0102)	-0.0177^{*} (0.0101)	-0.0144 (0.00903)	-0.00401 (0.00950)	-0.00393 (0.00947)	-0.00402 (0.00950)
$\triangle CashHoldings$		-0.0313^{***} (0.00811)	-0.0313^{***} (0.00812)	-0.0312^{***} (0.00814)		-0.0280^{***} (0.00768)	-0.0280^{***} (0.00767)	-0.0280^{***} (0.00766)
$\triangle CashHoldings_{t-1}$		$0.00696 \\ (0.00897)$	$0.00676 \\ (0.00899)$	$\begin{array}{c} 0.00708 \\ (0.00897) \end{array}$		-0.0100 (0.00720)	-0.00999 (0.00719)	-0.0100 (0.00721)
BOR_{t-1}			-0.00749^{*} (0.00447)				0.000428 (0.00110)	
$\triangle BOR2_{t-1}$				-0.00189 (0.00551)				$\begin{array}{c} 0.0000815 \\ (0.000892) \end{array}$
Observations	4810	3827	3827	3827	6403	5217	5217	5217
Firms	1008	949	949	949	1091	1054	1054	1054
No. of instruments	21	25	26	26	21	25	26	26
AR1 (p-value)	0.000	0.000	0.000	0.000	0.000	0.000	0.000217	0.000
AR2 (p-value)	0.731	0.360	0.367	0.363	0.098	0.174	0.175	0.175
Hansen-J (p-value)	0.765	0.579	0.572	0.582	0.009	0.0742	0.0741	0.0739
Standard errors in parentheses								

Table 5.2: Results for SMEs and Large Firms

* p<0.10, * * $p{<}0.05, * * * p{<}0.010$

The estimation results for large firms are reported in columns 5–8. The coefficient of lagged RD is significant at the 1 percent level, and are larger than those reported for MSMEs. As for MSMEs, lagged squared RD remains insignificant in all specifications. Lagged OpRevenue is insignificant for all specifications. The CashFlow coefficient decreases from a 1 percent significance level to 5 percent when the lagged and contemporaneous Δ CashHoldnings are included in the specification, with lower coefficients than in the sample with small firms. Lagged CashFlow is insignificant in all specifications. Δ CashHoldnings is negative and significant at the 1 percent, with negative coefficients that are slightly smaller in absolute value than for small firms. Lagged Δ CashHoldnings, BOR and Δ BOR2 are insignificant in all specifications.

5.3 Results for Listing Status

Table 5.3 presents the estimates for firms that are listed on the stock market and those that unlisted. The results for unlisted firms are displayed in columns 1–4. As has been the result of all specifications across the sample, lagged RD is significant at the 1 percent level in all specifications. Similarly to the results for MSMEs, the coefficients of lagged RD are relatively small and become smaller when controlling for lagged and contemporaneous Δ CashHoldnings.

	Unlisted				Listed				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
RD_{t-1}	$\begin{array}{c} 0.514^{***} \\ (0.139) \end{array}$	$\begin{array}{c} 0.477^{***} \\ (0.162) \end{array}$	$\begin{array}{c} 0.479^{***} \\ (0.161) \end{array}$	$\begin{array}{c} 0.476^{***} \\ (0.161) \end{array}$	0.901^{***} (0.142)	1.024^{***} (0.162)	1.007^{***} (0.145)	$\begin{array}{c} 1.007^{***} \\ (0.145) \end{array}$	1.009^{***} (0.144)
RD_{t-1}^2	0.201 (0.360)	$0.375 \\ (0.404)$	$0.365 \\ (0.398)$	$0.385 \\ (0.407)$	-0.282 (0.371)	-0.346 (0.638)	-0.437 (0.540)	-0.449 (0.538)	-0.448 (0.537)
$OpRevenue_{t-1}$	-0.00622^{**} (0.00305)	-0.00525 (0.00351)	-0.00497 (0.00351)	-0.00517 (0.00351)	-0.0125^{*} (0.00705)	-0.0109 (0.0157)	-0.0197 (0.0146)	-0.0188 (0.0146)	-0.0194 (0.0147)
CashFlow	$\begin{array}{c} 0.0572^{***} \\ (0.0111) \end{array}$	$\begin{array}{c} 0.0694^{***} \\ (0.0140) \end{array}$	0.0700^{***} (0.0140)	$\begin{array}{c} 0.0683^{***} \\ (0.0139) \end{array}$	$\begin{array}{c} 0.0339^{***} \\ (0.0125) \end{array}$	0.0343^{*} (0.0182)	0.0439^{**} (0.0171)	0.0436^{**} (0.0172)	0.0439^{**} (0.0170)
$\operatorname{CashFlow}_{t-1}$	-0.0217^{***} (0.00786)	-0.0178^{**} (0.00903)	-0.0176^{*} (0.00902)	-0.0182^{**} (0.00903)	-0.00659 (0.0125)	$\begin{array}{c} 0.0112\\ (0.0173) \end{array}$	$\begin{array}{c} 0.00970 \\ (0.0198) \end{array}$	0.00935 (0.0200)	0.00978 (0.0197)
$\triangle CashHoldings$		-0.0275^{***} (0.00645)	-0.0276^{***} (0.00645)	-0.0275^{***} (0.00645)			-0.0390^{**} (0.0161)	-0.0392^{**} (0.0161)	-0.0392^{**} (0.0161)
$\triangle \mathbf{CashHoldings}_{t-1}$		-0.00275 (0.00738)	-0.00284 (0.00740)	-0.00282 (0.00738)			$\begin{array}{c} 0.0121\\ (0.0166) \end{array}$	0.0114 (0.0165)	0.0117 (0.0166)
BOR_{t-1}			-0.00782^{**} (0.00346)					-0.000613 (0.00135)	
$\triangle BOR2_{t-1}$				-0.0126^{**} (0.00544)					0.000879 (0.00137)
StkIssues						-0.0397^{**} (0.0188)	-0.00613 (0.0264)	-0.00524 (0.0263)	-0.00525 (0.0267)
$StkIssues_{t-1}$						0.0291^{*} (0.0152)	0.0267 (0.0223)	0.0274 (0.0222)	0.0274 (0.0225)
Observations Firms No. of instruments AR1 (p-value) AR2 (p-value) Hansen-J (p-value)	6919 1457 21 6.71e-08 0.153 0.0590	$5378 \\ 1362 \\ 25 \\ 0.0000117 \\ 0.203 \\ 0.0908$	$5378 \\ 1362 \\ 26 \\ 0.000 \\ 0.202 \\ 0.0904$	$5378 \\ 1362 \\ 26 \\ 0.000 \\ 0.212 \\ 0.113$	$3943 \\ 577 \\ 21 \\ 0.000 \\ 0.638 \\ 0.615$	$\begin{array}{c} 2277\\ 476\\ 26\\ 0.00122\\ 0.526\\ 0.694 \end{array}$	$1963 \\ 464 \\ 30 \\ 0.005 \\ 0.788 \\ 0.719$	$1963 \\ 464 \\ 31 \\ 0.005 \\ 0.790 \\ 0.708$	$1963 \\ 464 \\ 31 \\ 0.005 \\ 0.792 \\ 0.715$

Table 5.3: Results for Unlisted and Listed Firms

Standard errors in parentheses

* p<0.10, * * p<0.05, * * *p<0.010

Furthermore, the lag of RD^2 is insignificant in all specifications for unlisted firms. Lagged OpRevenue is negative and significant at a 5 percent level in Column 1. CashFlow is significant at the 1 percent level in all specifications, with a coefficient that increases when controlling for lagged and contemporaneous Δ CashHoldnings. The coefficient of lagged CashFlow decreases in absolute value and significance when controlling for lagged and contemporaneous Δ CashHoldnings. When including BOR the significance level drops even further, from the 5 percent level to the 10 percent level. However, when controlling for Δ BOR2, it is still significant at the 5 percent level. Δ CashHoldnings is significant at the 1 percent level in all specifications where it is included, with negative coefficients. Lagged Δ CashHoldnings is insignificant in all specifications. Most importantly, both BOR and Δ BOR2 are significant at the 5 percent level for unlisted firms, with negative coefficients which are relatively small in absolute value.

The results for listed firms are displayed in columns 5–9. The specifications differ from previous specifications by also including lagged and contemporaneous estimations of StkIssues. The lagged RD variable is significant at the 1 percent level for all specifications, and the coefficients are larger than for listed firms (1.02 compared to 0.51 when only controlling for lagged and contemporaneous CashFlow). The lag of RD^2 is insignificant in all specifications. OpRevenue is negative and significant at the 10 percent level when including only cash flow, and is negative and non-significant in all other specifications. CashFlow is positive and significant at the 1 percent level in Column 5, and the coefficient becomes slightly larger in size but less significant when adding contemporaneous and lagged StkIssues. Lagged CashFlow is insignificant in all specifications. Contemporaneous StkIssues has a negative coefficient which is significant at the 5 percent level in Column 6, while lagged StkIssues is positive and significant at the 10 percent level. Both lagged and contemporaneous StkIssues are insignificant in all other specifications. Δ CashHoldnings is negative and significant at the 5 percent level in all specifications, with a coefficient that is slightly larger in absolute value when including BOR and $\Delta BOR2$. Lagged $\Delta BOR2$ is positive and insignificant for all specifications. Neither BOR nor $\Delta BOR2$ is significant in the sample with listed firms, and both coefficients are small in absolute values.

5.4 Analysis of the Results

The results reported in Section 5.1 are fairly consistent with previous literature on R&D financing. Across all specifications and categories, we find that CashFlow is significant and positive, indicating that R&D investment is subject to finance constraints. Similarly, we find that $\Delta CashHoldings$ is significant and negative. This result indicates that firms smooth their R&D expenditures, because of reductions in cash holdings allowing liquidity to be available for R&D expenditure. In addition, we find that the lag of RD is positive and significant, illustrating that R&D expenditures are persistent and once again indicating expenditure smoothing. Furthermore, we find three indications that MSMEs and unlisted firms are more financially constrained than large and listed firms. First, the coefficients of CashFlow for are higher for unlisted firms than listed firms, and for MSMEs they are almost twice the size of corresponding coefficients of large firms. Second, the coefficients of Δ CashHoldings are larger in absolute terms for MSMEs and unlisted firms than for large and listed firms. These results indicate financial constraints, implying that these firms need to fund a higher proportion of their R&D expenditures with internal funds compared to listed and large firms. Third, the coefficients of lagged RD are higher for large and listed firms than for MSMEs and unlisted firms. A plausible interpretation of this result is that R&D expenditures are less persistent for these firms because they are unable to smooth spending as well as large and listed firms – once again indicating that they are subject to more stringent financial constraints.

As for our main contribution – the addition of BOR and Δ BOR2 – it is evident that the results vary depending on which category is being considered. When interpreting the results of Table 5.1 containing all firms in the sample, neither the BOR nor the Δ BOR2 estimates is statistically significant. This result is to be expected because of the heterogeneity in the sample. As mentioned in section 2.3, we expect that the implementation of macroprudential policies will primarily affect the most financially constrained firms, which our theory and our results indicate are unlisted firms and MSMEs. The effects of macroprudential policies are not expected to be prevalent in large and listed firms with strong internal funds and/or better access to external sources of finance. Since these firms constitute a large part of the whole sample, the results of Table 5.1 are consistent with our theory. Our theory is further supported by the results of Table 5.3, which present the results for listed and unlisted firms, and Table 5.2 presenting results for MSMEs and large firms. While the coefficients of BOR and Δ BOR2 are insignificant for large and listed firms, both are negative and significant at the 5 percent level for unlisted firms. Similarly, Δ BOR2 is negative and significant at the 10 percent level for MSMEs. However, the significant coefficients are small in absolute value relative to other variables. This result illustrates that R&D expenditure is primarily financed with internal funds, which is consistent with previous literature.

A somewhat unexpected result is that $\Delta BOR2$ is significant for MSMEs but not BOR. The interpretation is that a tightening of macroprudential policies significantly reduces R&D expenditures but not an implementation, which could seem contradictory. Further evidence of this can be found in the results for unlisted firms, where the coefficient of $\Delta BOR2$ is larger in absolute terms than that of BOR. A plausible explanation for this is that implementations of macroprudential policies are also coded as tightening measures in $\Delta BOR2$, provided that they occurred during the sample period. Therefore, the coefficient of $\Delta BOR2$ reflects the effect of both an implementation and a subsequent tightening. Another possibility is that the effect of going from 0 to 1 instrument in place is large, but that the marginal effect of adding one more is small. This situation would reduce the size of the coefficient for BOR.

5.4.1 Implications of the Results

Our results indicate that implementation and tightening of borrower-targeted macroprudential policies are associated with lower R&D expenditures for unlisted firms and to some degree for MSMEs. The main implication of this result is that such policies could have an adverse effect on economic growth, especially in countries with a large degree of MSMEs or with less developed stock markets. In other words, our results provide further evidence of a trade-off between stability and growth in the use of macroprudential policies, as suggested by Sánchez and Röhn (2016). Another implication is that unlisted firms and MSMEs are more dependent on bank loans to finance their activities than listed and large firms, and that they may use debt directly for their R&D investments.

Because the results suggest a trade-off, their inference could be used to formulate policies, depending on policy objectives. The aim of macroprudential policies is to increase finan-

cial stability, but knowing that this can come at the expense of economic growth suggests that cost-benefit analyses have relevance. Furthermore, it suggests that the choice of policy instrument is important. Thus far, there is only evidence that borrower-targeted macroprudential policies are associated with lower R&D expenditures and firm investments. Since evidence suggests that lender-targeted policies are effective in increasing stability but have not been shown to have the same adverse effects, use of these may be advantageous if the goal is to minimize the effects on firm investment. However, the results should be cautiously interpreted when discussing the effects of macroprudential policies. The aim of this study has not been to argue for fewer or less intense borrower-targeted policies but to provide evidence of the externalities they have on R&D expenditures. This study has only provided preliminary evidence that borrower-targeted macroprudential policies are associated with lower R&D expenditures. Though we have presented two possible explanations of how this may occur, we have not proven either of these causalities. Furthermore, the results are based on aggregated observations and can not explain the effects of individual policies in individual countries. Further research is needed to confirm and expand upon these findings.

5.5 Robustness Checks

5.5.1 Comparison to Brown et al. (2012)

When estimating the coefficients for large firms in our sample, the results differ from those of Brown et al. (2012), even though the baseline specifications and models are similar. This disparity can be explained by noting the difference in samples between the studie. Comparing our summary statistics in Table 3.1 to those in Brown et al., it is clear we have a different firm composition. The mean and median of employees in Brown et al.'s full sample is 10,607 and 1,570 respectively, which shows the firms in our sample are of considerably smaller size. Because Brown et al. only includes firms listed on the stock market, the mean of net proceeds from stock issues (0.108) is also higher than in our sample. This pattern can be observed throughout the rest of the comparison, with the variables in Brown et al. (2012) having slightly larger mean and median than the variables used in this study. Furthermore, Brown et al. defined large firms as those that were above the 70th percentile in terms of employees in their sample, not as those with more than 250 employees. As a robustness check to make sure that the model is correctly specified, the model has been estimated using a similar size definition, instrument specification and listing status as Brown et al. Since our sample contains a lower median of employees, we defined large firms as those above the 90th percentile in terms of employees in our sample. Furthermore, only firms listed on the stock exchange are included. Comparing results to Brown et al.'s sample of large firms across Europe, we obtain similar results.¹⁰ As in Brown et al. (2012), the lagged value of RD is large and positive, while the squared lag of RD is negative and large in absolute value. Furthermore, the coefficients of OpRev (sales in Brown et al. (2012)), StkIssues and sum of CashFlow and Δ CashHoldings variables are consistent with their results in both coefficients and level of significance. Therefore, our results imply that the baseline regression is correctly specified according to previous research.

5.5.2 Weighting the Sample

Because the model is estimated at an aggregate level and there is an uneven distribution of firms across our sample of countries, and some countries are over-represented in the sample. Since we expect macroprudential policies to have different effects in different countries, this over-representation could potentially skew the results. To control for this aspect, we estimate the model after inverse-weighting the sample. Each category (size and listing status) has been estimated with the full set of applicable financial variables, including BOR and Δ BOR2 respectively. Results including BOR are presented in Table 5.4 and results including Δ BOR2 are presented in Table 5.5.

¹⁰The estimation results can be found in Table 7.1 in the Appendix.

	S	ME	La	arge	Un	listed	Listed	
	Weighted	Unweighted	Weighted	Unweighted	Weighted	Unweighted	Weighted	Unweighted
RD_{t-1}	0.502***	0.560***	0.893***	0.898***	0.529***	0.479***	0.965***	1.007***
	(0.154)	(0.153)	(0.190)	(0.170)	(0.177)	(0.161)	(0.127)	(0.145)
RD_{t-1}^2	0.417	0.243	-0.591	-0.577	0.329	0.365	-0.405	-0.449
v 1	(0.317)	(0.308)	(0.390)	(0.431)	(0.411)	(0.398)	(0.414)	(0.538)
$OpRevenue_{t-1}$	-0.00174	-0.00593*	-0.00140	-0.00301	-0.00346	-0.00497	-0.0184	-0.0188
	(0.00452)	(0.00353)	(0.00673)	(0.00503)	(0.00448)	(0.00351)	(0.0150)	(0.0146)
BOR_{t-1}	-0.0114*	-0.00749*	0.000439	0.000428	-0.00654	-0.00782**	-0.000717	-0.000613
	(0.00670)	(0.00447)	(0.00147)	(0.00110)	(0.00560)	(0.00346)	(0.00156)	(0.00135)
CashFlow	0.0645***	0.0612***	0.0444**	0.0330**	0.0636***	0.0700***	0.0376	0.0436**
	(0.0131)	(0.0124)	(0.0190)	(0.0129)	(0.0159)	(0.0140)	(0.0232)	(0.0172)
$CashFlow_{t-1}$	-0.0146	-0.0173*	0.00453	-0.00393	-0.0200*	-0.0176*	0.00915	0.00935
	(0.0106)	(0.0102)	(0.0128)	(0.00947)	(0.0108)	(0.00902)	(0.0213)	(0.0200)
△CashHoldings	-0.0370***	-0.0313***	-0.0312***	-0.0280***	-0.0292***	-0.0276***	-0.0426*	-0.0392**
	(0.00845)	(0.00812)	(0.00899)	(0.00767)	(0.00726)	(0.00645)	(0.0228)	(0.0161)
$\triangle CashHoldings_{t-1}$	0.00293	0.00676	-0.0130	-0.00999	-0.00113	-0.00284	0.0162	0.0114
	(0.00998)	(0.00899)	(0.00850)	(0.00719)	(0.00770)	(0.00740)	(0.0187)	(0.0165)
StkIssues							-0.00962	-0.00524
							(0.0227)	(0.0263)
$StkIssues_{t-1}$							0.0355**	0.0274
							(0.0159)	(0.0222)
Observations	3827	3827	5217	5217	5378	5378	1963	1963
Firms	949	949	1054	1054	1362	1362	464	464
No. of instruments	26	26	26	26	26	26	31	31
AR1 (p-value)	0.000	0.000	0.001	0.000	0.000	0.000	0.006	0.005
AR2 (p-value)	0.363	0.367	0.118	0.175	0.356	0.202	0.816	0.790
Hansen-J (p-value)	0.692	0.572	0.204	0.0741	0.0375	0.0904	0.717	0.708
Standard arrors in parantheses								

Table 5.4: Comparison of Weighted and Unweighted Results, BOR1

* p<0.10, * * p<0.05, * * *p<0.010

As illustrated in Table 5.4, results are similar when comparing weighted and unweighted samples. Considering MSMEs, OpRevenue ceases to be significant when weighting the sample. However, the coefficient is still small in absolute value. Furthermore, the Cash-Flow in the listed firm sample is no longer significant when the sample is weighted. The coefficient of Δ CashHoldings becomes larger in absolute value but less significant in the weighted sample for listed firms, while StkIssues becomes significant at the 5 percent level. The lag of BOR is not significant in the weighted sample for unlisted firms, compared to being significant at the 5 percent level in the unweighted sample. A possible explanation for this result, that is consistent with our theory, is that countries that have gained more weight in the sample are more market-based. As discussed in Section 2.3, we suspect firms in these countries to be less dependent on bank loans therefore be less affected by implementation of borrower-targeted policies. Another possible explanation is that countries that have more weight in the sample contribute with unlisted firms that are large. Since large firms are expected to be less financially constrained, they should also be less affected by implementation of borrower-targeted policies.

	SI	ME	La	arge	Unl	listed	L	isted
	Weighted	Unweighted	Weighted	Unweighted	Weighted	Unweighted	Weighted	Unweighted
RD_{t-1}	0.500^{***} (0.154)	0.560^{***} (0.152)	0.892^{***} (0.190)	0.896^{***} (0.168)	$\begin{array}{c} 0.518^{***} \\ (0.179) \end{array}$	0.476^{***} (0.161)	0.968^{***} (0.125)	1.009^{***} (0.144)
RD_{t-1}^2	$\begin{array}{c} 0.426 \\ (0.319) \end{array}$	$\begin{array}{c} 0.246 \\ (0.310) \end{array}$	-0.590 (0.390)	-0.573 (0.430)	$\begin{array}{c} 0.353 \\ (0.418) \end{array}$	$0.385 \\ (0.407)$	-0.404 (0.411)	-0.448 (0.537)
$OpRevenue_{t-1}$	-0.00190 (0.00452)	-0.00597^{*} (0.00355)	-0.00132 (0.00691)	-0.00293 (0.00502)	-0.00352 (0.00449)	-0.00517 (0.00351)	-0.0193 (0.0156)	-0.0194 (0.0147)
$\triangle BOR2_{t-1}$	-0.00140 (0.00610)	-0.00189 (0.00551)	$\begin{array}{c} 0.000394 \\ (0.00111) \end{array}$	$\begin{array}{c} 0.0000815 \\ (0.000892) \end{array}$	-0.0114^{**} (0.00558)	-0.0126^{**} (0.00544)	$\begin{array}{c} 0.000456 \\ (0.00122) \end{array}$	$\begin{array}{c} 0.000879 \\ (0.00137) \end{array}$
CashFlow	0.0648^{***} (0.0132)	$\begin{array}{c} 0.0611^{***} \\ (0.0124) \end{array}$	0.0447^{**} (0.0189)	0.0331^{**} (0.0129)	$\begin{array}{c} 0.0628^{***} \\ (0.0159) \end{array}$	$\begin{array}{c} 0.0683^{***} \\ (0.0139) \end{array}$	$\begin{array}{c} 0.0371 \\ (0.0231) \end{array}$	0.0439^{**} (0.0170)
$CashFlow_{t-1}$	-0.0148 (0.0105)	-0.0177^{*} (0.0101)	0.00449 (0.0129)	-0.00402 (0.00950)	-0.0206^{*} (0.0109)	-0.0182^{**} (0.00903)	0.00981 (0.0212)	0.00978 (0.0197)
\triangle CashHoldings	-0.0369*** (0.00842)	-0.0312^{***} (0.00814)	-0.0311^{***} (0.00897)	-0.0280^{***} (0.00766)	-0.0292^{***} (0.00727)	-0.0275^{***} (0.00645)	-0.0421* (0.0228)	-0.0392^{**} (0.0161)
$\triangle CashHoldings_{t-1}$	0.00308 (0.00999)	0.00708 (0.00897)	-0.0132 (0.00846)	-0.0100 (0.00721)	-0.00149 (0.00774)	-0.00282 (0.00738)	0.0164 (0.0188)	0.0117 (0.0166)
StkIssues							-0.0101 (0.0232)	-0.00525 (0.0267)
$StkIssues_{t-1}$							0.0354^{**} (0.0161)	0.0274 (0.0225)
Observations Firms	3827 949 26	3827 949 26	5217 1054 26	5217 1054 26	5378 1362 26	5378 1362 26	1963 464 31	$1963 \\ 464 \\ 31$
AR1 (p-value) AR2 (p-value)	$0.000 \\ 0.357$	0.000 0.363	0.001 0.118	$0.000 \\ 0.175$	0.000 0.368	0.000 0.212	0.006 0.817	0.005 0.792

Table 5.5: Comparison of Weighted and Unweighted Results, BOR2

standard errors in parentneses

* p<0.10, * * $p{<}0.05, * * * p{<}0.010$

Table 5.5 displays a comparison on weighted and unweighted results when controlling for Δ BOR2. As is the case in Table 5.4, weighted and unweighted results are very similar. When considering this specification, the lagged difference of Δ BOR2 continues to be significant at the 5 percent level with a coefficient marginally lower in absolute value. That Δ BOR2 continues to be significant after weighting but BOR does not further supports our previous results that indicate that a tightening of borrower-targeted instruments have more effect than an implementation.

6 Conclusion

This study has analyzed the relationship between borrower-targeted macroprudential policies and firm spending on R&D. Using a system GMM approach on a sample of European firms, we have provided preliminary evidence that implementation and tightening of such policies are associated with lower R&D spending for unlisted firms and to some degree for MSMEs. However, the effect of borrower-targeted policies on R&D is small compared to the effects of other financial variables such as cash holdings and cash flow. This result is consistent with previous research, which suggests that R&D is primarily financed using internal funds.

We have proposed that unlisted firms and MSMEs are affected because they are more financially constrained than large and listed firms, and that these constraints become more stringent after implementation and tightening of borrower-targeted policies. Furthermore, we have suggested that the causality between borrower-targeted policies and reduced R&D expenditures can either be direct or indirect. If direct, firms use bank loans to some degree in order to fund their R&D expenditures. If indirect, firms use internal funds to finance R&D and use bank loans to fund other activities which become subject to finance constraints after implementation and tightening of policies. Firms will then need to substitute their use of bank funds for those activities with internal funds. Since these are limited, firms will need to prioritize their use of them and possibly choose to reduce R&D expenditure.

To the best of our knowledge, this is the first study to investigate the relationship between R&D expenditure and macroprudential policies. Since our results indicate a relationship between the two, further research on the topic is of interest. For example, we have been unable to investigate the difference between bank– and market–based economies. Further research can therefore focus on specific countries or investigate how the effects differ based on the size and quality of financial institutions. Furthermore, we have not provided conclusive evidence of either of the proposed causalities mentioned above. Another suggestion for further research is therefore to examine the R&D financing decision compared to other types of investment and include controls for external debt. Such a study could empirically investigate if the effect of borrower-targeted policies on R&D investment is indeed an indirect effect of it being deproioritized compared to less risky investments.

References

- Aghion, P., Bond, S., Klemm, A. and Marinescu, I. (2004), 'Technology and Financial Structure: Are Innovative Firms Different?', Journal of the European Economic Association 2(2-3), 277–288. URL: https://doi.org/10.1162/154247604323067989
- Aghion, P., Farhi, E. and Kharroubi, E. (2012), Monetary policy, liquidity, and growth, NBER Working Papers 18072, National Bureau of Economic Research, Inc. URL: https://EconPapers.repec.org/RePEc:nbr:nberwo:18072
- Alam, Z., Nier, E., Kang, M. H., Narita, M. M., Alter, M. A., Wang, N., Eiseman, J. and Gelos, M. R. G. (2019), Digging Deeper–Evidence on the Effects of Macroprudential Policies from a New Database, IMF Working Papers 2019/066, International Monetary Fund.

URL: https://ideas.repec.org/p/imf/imfwpa/2019-066.html

Arellano, M. and Bond, S. (1991), 'Some tests of specification for panel data: Monte carlo evidence and an application to employment equations', *Review of Economic Studies* 58(2), 277–297.

URL: https://EconPapers.repec.org/RePEc:oup:restud:v:58:y:1991:i:2:p:277-297.

- Arellano, M. and Bover, O. (1995), 'Another look at the instrumental variable estimation of error-components models', Journal of Econometrics 68(1), 29–51.
 URL: https://www.sciencedirect.com/science/article/pii/030440769401642D
- Ayyagari, M., Soledad, M., Peria, M. and Beck, T. (2018), The Micro Impact of Macroprudential Policies: Firm-Level Evidence, IMF Working Papers 2018/267, International Monetary Fund.

URL: https://ideas.repec.org/p/imf/imfwpa/2018-267.html

- Bhagat, S. and Welch, I. (1995), 'Corporate research development investments international comparisons', Journal of Accounting and Economics 19(2-3), 443–470.
 URL: https://EconPapers.repec.org/RePEc:eee:jaecon:v:19:y:1995:i:2-3:p:443-470
- Blundell, R. and Bond, S. (1998), 'Initial conditions and moment restrictions in dynamic panel data models', Journal of Econometrics 87(1), 115–143.
 URL: https://ideas.repec.org/a/eee/econom/v87y1998i1p115-143.html

Boar, C., Gambacorta, L., Lombardo, G. and Pereira da Silva, L. A. (2017), 'What are the effects of macroprudential policies on macroeconomic performance?', BIS Quarterly Review.

URL: *https://EconPapers.repec.org/RePEc:bis:bisqtr:1709g*

- Bond, S., Harhoff, D. and Reenen, J. V. (2005), 'Investment, rd and financial constraints in britain and germany', Annales d'Économie et de Statistique (79/80), 433–460.
 URL: http://www.jstor.org/stable/20777584
- Bond, S. and Meghir, C. (1994), 'Financial constraints and company investment', Fiscal Studies 15(2), 1–18.

URL: https://EconPapers.repec.org/RePEc:ifs:fistud:v:15:y:1994:i:2:p:1-18

- Brown, J., Martinsson, G. and Petersen, B. (2012), 'Do financing constraints matter for rd?', European Economic Review 56(8), 1512–1529.
 URL: https://EconPapers.repec.org/RePEc:eee:eecrev:v:56:y:2012:i:8:p:1512-1529
- Cerutti, E., Claessens, S. and Laeven, L. (2017), 'The use and effectiveness of macroprudential policies: New evidence', Journal of Financial Stability 28, 203–224.
 URL: https://www.sciencedirect.com/science/article/pii/S1572308915001035
- Czarnitzki, D. and Hottenrott, H. (2009), 'Rd investment and financing constraints of small and medium-sized firms', *Small Business Economics* 36, 65–83.
- Fazzari, S., Hubbard, R. and Petersen, B. (1988), 'Financing constraints and corporate investment', National Bureau of Economic Research, Inc, NBER Working Papers 1.
- Galati, G. and Moessner, R. (2013), 'Macroprudential policy a literature review', Journal of Economic Surveys 27(5), 846–878.
 - **URL:** https://onlinelibrary.wiley.com/doi/abs/10.1111/j.1467-6419.2012.00729.x
- Hall, B. H. (1992), Investment and Research and Development at the Firm Level: Does the Source of Financing Matter?, NBER Working Papers 4096, National Bureau of Economic Research, Inc.

URL: *https://ideas.repec.org/p/nbr/nberwo/4096.html*

Hall, B. H. and Lerner, J. (2010), Chapter 14 - the financing of rd and innovation, inB. H. Hall and N. Rosenberg, eds, 'Handbook of The Economics of Innovation, Vol. 1',

Vol. 1 of Handbook of the Economics of Innovation, North-Holland, pp. 609–639.
URL: https://www.sciencedirect.com/science/article/pii/S0169721810010142

- Harhoff, D. (1998), 'Rd and productivity in german manufacturing firms', Economics of Innovation and New Technology 6(1), 29–50.
 URL: https://EconPapers.repec.org/RePEc:taf:ecinnt:v:6:y:1998:i:1:p:29-50
- Howe, J. and McFetridge, D. (1976), 'The determinants of r d expenditures', The Canadian Journal of Economics / Revue canadienne d'Economique 9(1), 57–71.
 URL: http://www.jstor.org/stable/134415
- Lim, C., Costa, A., Columba, F., Kongsamut, P., Otani, A., Saiyid, M., Wezel, T. and Wu, X. (2011), 'Macroprudential policy: What instruments and how to use them? lessons from country experiences', SSRN Electronic Journal.
- Lucey, B. and Bhaird, C. (2006), 'Capital structure and the financing of smes: Empirical evidence from an irish survey', *SSRN Electronic Journal*.
- Magri, S. (2014), Does issuing equities help R&D activity? Evidence from unlisted Italian high-tech manufacturing firms, Temi di discussione (Economic working papers) 978, Bank of Italy, Economic Research and International Relations Area.
 URL: https://ideas.repec.org/p/bdi/wptemi/td₉78₁4.html
- Mulkay, B., Hall, B. H. and Mairesse, J. (2001), Firm Level Investment and R&D in France and the United States: A Comparison, Economics Papers 2001-W2, Economics Group, Nuffield College, University of Oxford. URL: https://ideas.repec.org/p/nuf/econwp/0102.html
- Müller, E. and Zimmermann, V. (2006), The importance of equity finance for rd activity – are there differences between young and old companies?, Discussion paper series of sfb/tr 15 governance and the efficiency of economic systems, Free University of Berlin, Humboldt University of Berlin, University of Bonn, University of Mannheim, University of Munich.

URL: *https://EconPapers.repec.org/RePEc:trf:wpaper:111*

Orbis (2022), 'Orbis database'.

 $\label{eq:urrel} \textbf{URL:} https://www.bvdinfo.com/en-gb/solutions-for-your-role/business-development-and-strategy?gclid=Cj0KCQjwhLKUBhDiARIsAMaTLnGDatMyVbslMx6FrKpbs4YL2YaHrquDUbbh$

- Roodman, D. (2009), 'How to do xtabond2: An introduction to difference and system GMM in Stata', Stata Journal 9(1), 86–136.
 URL: https://ideas.repec.org/a/tsj/stataj/v9y2009i1p86-136.html
- Sánchez, A. C. and Röhn, O. (2016), 'How do policies influence gdp tail risks?', (1339). URL: https://www.oecd-ilibrary.org/content/paper/5jln0428l1wl-en
- Ughetto, E. (2008), 'Does internal finance matter for rd? new evidence from a panel of italian firms', Cambridge Journal of Economics 32(6), 907–925. URL: http://www.jstor.org/stable/23602044
- Wang, H., Liang, P., Li, H. and Yang, R. (2016), 'Financing sources, rd investment and enterprise risk', *Procedia Computer Science* 91, 122–130. Promoting Business Analytics and Quantitative Management of Technology: 4th International Conference on Information Technology and Quantitative Management (ITQM 2016).
 - URL: https://www.sciencedirect.com/science/article/pii/S1877050916312303

7 Appendix

	(1)	(2)	(3)
RD_{t-1}	$\begin{array}{c} 0.998^{***} \\ (0.204) \end{array}$	$\begin{array}{c} 1.139^{***} \\ (0.158) \end{array}$	$ \begin{array}{c} 1.438^{***} \\ (0.311) \end{array} $
RD_{t-1}^2	-0.978^{**} (0.443)	-1.367^{***} (0.292)	-1.721^{***} (0.597)
$OpRevenue_{t-1}$	$0.00206 \\ (0.00559)$	0.00242 (0.0121)	-0.0186 (0.0153)
CashFlow	$0.0165 \\ (0.0158)$	0.0502 (0.0539)	$0.0797 \\ (0.0509)$
$\operatorname{CashFlow}_{t-1}$	0.0127 (0.0127)	0.00434 (0.0239)	0.0468 (0.0311)
StkIssues		0.0412 (0.0715)	0.0728 (0.0608)
$StkIssues_{t-1}$		-0.0126 (0.0266)	-0.0195 (0.0640)
$\triangle CashHoldings$			-0.0426 (0.0335)
$\triangle CashHoldings_{t-1}$			-0.0395^{*} (0.0227)
Observations	1560	984	850
Firms	226	196	192
No. of instruments	21	26	30
AR1 (p-value)	0.0136	0.0563	0.0250
AR2 (p-value)	0.993	0.980	0.599
Hansen-J (p-value)	0.539	0.0787	0.0792

Table 7.1: Specification of Brown et al.

Standard errors in parentheses

* p<0.10, * * $p{<}0.05, * * * p{<}0.010$

Country	Firms
AT	22
BE	20
DE	233
DK	32
ES	8
FI	42
\mathbf{FR}	75
GB	1296
GR	18
IE	23
IT	8
LU	10
NL	24
SE	266
Total	2078

 Table 7.2: Countries Included in the Full Sample